Author's response to reviews

Title: A large left ventricular pseudoaneurysm in a case of Behcet`s disease and literature review

Authors:

Seyed-Mojtaba Marashi (marashi@doctor.com)
Payam Eghtesadi-Araghi (payam_eghtesadi@yahoo.com)
Mohammad-hussein Mandegar (mhmandegar@yahoo.com)

Version: 3 Date: 28 May 2005

Author's response to reviews:

Dear Dr Moylan, May 24, 2005

Please accept our appreciation. It is a pleasure that our manuscript was published in your renowned journal. We are also pleased that your reviewers have found our article important in its field. We hope your readers would also find it helpful.

We have revised our manuscript considering your formatting comments point-by-point which follows. You can find all these alterations in the manuscript. Thanks again.

Sincerely
Seyed Mojtaba Marashi, [Corresponding author]
Assistant Professor of Anesthesiology

1. We changed our title to ‘A large left ventricular pseudoaneurysm in Behcet’s disease: a case report’.
2. We removed titles from the list of authors.
3. We corrected 'IRAN' to 'Iran' in each case.
4. Abstract was structured into:
   a. Background,
   b. Case Presentation,
   c. Conclusions.
5. References were cited in the text using consecutive numbers in square brackets.
6. 'Discussion' was changed to 'Conclusions'?
7. 'The authors declare that they have no competing interests' was declared in competing interests section.
8. We added 'Mohammad Hussein Mandegar' to the authors list and consequently omitted him from the 'acknowledgement' section.